\begin{table}%t3
\caption{Precession solutions for $\iota = 5$--$8\degr$.}
\label{tb:sol-iota}
%\centerline
{
\begin{tabular}{c|ccccc}
\hline\hline
Parameter & \multicolumn{5}{c}{Value} \\
\hline
$\iota$ $[\degr]$ & 5 & 6 & {\bf 6.5} & 7 & 8 \\
$\Omega_{\rm p}$ $[\degr]$ & 1.3 & 1.4 & {\bf 1.45} & 1.5 & 1.6 \\
$t_{{\rm beam}}$ $[{\rm yr}]$ & 220 & 200 & {\bf 180} & 165 & 150 \\
$T_{\rm p}$ $[{\rm yr}]$ & 3320 & 2780 & {\bf 2570} & 2370 & 2070 \\ 
\hline
\end{tabular}}
\medskip
Note: Bold face denotes the best fit solution. 
$T_{\rm p}$ is calculated with the best fit value of the Alfv\'en~speed $V_{\rm A} = 0.078~c$. 
\end{table}